Author’s response to reviews

Title: Routine health information system utilization and factors associated thereof among health workers at government health institutions in East Gojjam Zone, Northwest Ethiopia

Authors:

Atsede Shiferaw (atsede.mazengia@yahoo.com)
Dessalegn Tegabu (zdesalegn@gmail.com)
Solomon Assefa (solomonazezo@gmail.com)
Melaku Yenit (melaku98@gmail.com)

Version: 2 Date: 01 Jul 2017

Author’s response to reviews:

Dear reviewers and editors,

Thank you very much for your critical contribution!

Limitation section is now included in the discussion section on page 11 lines 3-8

Policy context on health information system is included in the revised version on introduction page 4 line 12-28 and also on discussion section.